Author's response to reviews

Title: Field test of a novel detection device for Mycobacterium tuberculosis antigen in cough.

Authors:

Ruth McNerney (Ruth.Mcnerney@lshtm.ac.uk)
Beyene A Wondafrash (beyenewondafrash@googlemail.com)
Kebede Amena (kebedeamena@yahoo.com)
Ato Tesfaye (none@lshtm.ac.uk)
Elaine M McCash (elaine.mccash@rapidbiosensor.com)
Nicol J Murray (dr.nicol.murray@rapidbiosensor.com)

Version: 6 Date: 3 June 2010

Author's response to reviews:

Formatting changes to be made:

Major revisions

Manuscript: this manuscript does not have the correct structure. Please reformat it to use these headings:

Abstract, Background, Results, Discussion, Conclusions, Methods, Authors' contributions, Authors' information (optional), Acknowledgements (optional), References

We have adjusted the structure as requested.

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution.

We have added the full details for the authors based in Ethiopia. They are senior staff in this small hospital and are not ascribed to a particular department.

Authors' contributions: Please indicate that all authors read and approved the final manuscript. A sentence has been added as requested.

Minor revisions

Highlighting: Please remove.

The highlighting has been removed.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the
final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

We have undertaken a final check.